Author’s response to reviews

Title: MR imaging features of benign retroperitoneal paragangliomas and schwannomas

Authors:
yanguang shen (shenyanguang@163.com)
Yan Zhong (c302643187c@163.com)
Haiyi Wang (302643187@qq.com)
Lu Ma (b302643187b@163.com)
Yingwei Wang (a32643187a@163.com)
Kun Zhang (bbb7691188@qq.com)
Zhonghua Sun (Z.Sun@exchange.curtin.edu.au)
Huiyi Ye (13701100368@163.com)

Version: 3 Date: 29 Nov 2017

Author’s response to reviews:

Dear Editor,

Re: NURL-D-17-00108R

MR imaging features of benign retroperitoneal paragangliomas and schwannomas

We thank the editors for their constructive comments and suggestions to improve our manuscript. Please find enclosed a revised manuscript with this covering letter documenting all actions as requested by the editors.

Editor Comments:

1. Author’s Contributions:
a. We noted that you have included initials (JP, Jingjing Pan) that do not correspond to any of the authors currently listed on the title page nor the submission system. Please could you clarify this?
b. YS appears twice for the same role. Please correct this.

Response: Thank you for comment. Jingjing Pan (JP) was assistance with the preparation of this manuscript with authors. She has been acknowledged in the ‘Acknowledgements’ section (Line 367-369). We have deleted JP in the ‘Author Contributions’ section. We have revised the ‘YS’ (Line 379--382).

2. Acknowledgements:
   a. We note that you have not included an acknowledgements section. If you have no acknowledgements please state ‘Not Applicable’ in this section.
   b. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If no acknowledgements are necessary then please state 'None'.
Response: We have added the Acknowledgements section in the manuscript(Line 367-369).

3. Ethics Approval:
Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.
Response: We have added the the full name of the ethics committee in the manuscript (Line 387-389).

4. Declarations:
Please include a ‘Declarations’ heading before the funding information.
Response: We have added a ‘Declarations’ heading before the funding information (Line 366).